Author’s response to reviews

Title: Correlational analysis between neutrophil granulocyte levels and osteonecrosis of the femoral head

Authors:

Jiang jiang (double-jiang@qq.com)
xuqiang Liu (shliuxuqiang@163.com)
baojian Lai (LAIBAOJ@163.com)
dengjiong Hu (1501114570@qq.com)
lizhen Lai (739411295@qq.com)
jiaxiang Xu (605212509@qq.com)
songqing Chen (1132618334@qq.com)
Xiaofeng Li (Doublejiang9@163.com)

Version: 4 Date: 20 Aug 2019

Author’s response to reviews:

Editor-in-Chief
BMC Musculoskeletal Disorders

Dear Editor:

Thank you for reviewing our manuscript entitled “Correlational analysis between neutrophil granulocyte levels and osteonecrosis of the femoral head” (Ref.: No. BMSD-D-19-00397R3)”. We thank the reviewers for their positive comments and appreciate their valuable suggestions, and have now revised the manuscript accordingly to address the concerns raised by the reviewers.